



Article Locking-Enabled Security Analysis of Cryptographic Circuits ⁺

Devanshi Upadhyaya *, Maël Gay and Ilia Polian

Institut für Technische Informatik, Universität Stuttgart, Pfaffenwaldring 47, 70569 Stuttgart, Germany; mael.gay@iti.uni-stuttgart.de (M.G.); ilia.polian@iti.uni-stuttgart.de (I.P.)

- $* \ Correspondence: devanshi.upadhyaya@iti.uni-stuttgart.de$
- [†] This paper is an extended version of our paper published in the IEEE International Symposium on Hardware Oriented Security and Trust (HOST) in San Jose, CA, USA, 1–4 May 2023.

Abstract: Hardware implementations of cryptographic primitives require protection against physical attacks and supply chain threats. This raises the question of *secure composability* of different attack countermeasures, i.e., whether protecting a circuit against one threat can make it more vulnerable against a different threat. In this article, we study the consequences of applying logic locking, a popular design-for-trust solution against intellectual property piracy and overproduction, to cryptographic circuits. We show that the ability to unlock the circuit incorrectly gives the adversary new powerful attack options. We introduce LEDFA (locking-enabled differential fault analysis) and demonstrate for several ciphers and families of locking schemes that fault attacks become possible (or consistently easier) for incorrectly unlocked circuits. In several cases, logic locking has made circuit implementations prone to classical algebraic attacks with no fault injection needed altogether. We refer to this "zero-fault" version of LEDFA by the term LEDA, investigate its success factors in-depth and propose a countermeasure to protect the logic-locked implementations against LEDA. We also perform test vector leakage assessment (TVLA) of incorrectly unlocked AES implementations to show the effects of logic locking regarding side-channel leakage. Our results indicate that logic locking is not safe to use in cryptographic circuits, making them less rather than more secure.

Keywords: cryptography; logic locking; fault attacks; side channel analysis

Citation: Upadhyaya, D.; Gay, M.; Polian, I. Locking-Enabled Security Analysis of Cryptographic Circuits. *Cryptography* **2024**, *8*, 2. https:// doi.org/10.3390/cryptography8010002

check for updates

Academic Editor: Jim Plusquellic

Received: 7 November 2023 Revised: 29 November 2023 Accepted: 4 December 2023 Published: 5 January 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

In recent years, the security of sensitive data has become increasingly important, especially for embedded devices or Internet of Things (IoT) applications. It is crucial to ensure privacy and integrity while meeting hardware constraints. For instance, smartphones process a growing amount of sensitive data, and it is the responsibility of circuit designers to guarantee data integrity and prevent leakage. As a result, it has been widely accepted that hardware is an essential component of any critical electronic system and must be secure.

Data security can, to some extent, be improved by using cryptographic primitives, but physical limitations in power and space need to be considered as well [1]. Striking a balance between hardware constraints and security is crucial. However, it may lead to design flaws that an attacker can take advantage of. Therefore, implementation must be carefully planned, especially if the attacker has physical access to the device. An attacker can study the device and use various physical attacks to achieve their objectives.

Several attack vectors have been identified that specifically target hardware implementations at various stages of a circuit's life cycle and aim to compromise different assets [2]. Therefore, various countermeasures have been proposed against specific attack vectors, such as side-channel analysis, fault-injection attacks, hardware manipulation, or counterfeiting. However, it has been observed that these countermeasures are not readily composable [3]. A countermeasure that is effective against one type of attack may actually increase the vulnerability of the circuit to other attacks. For instance, in [4], adding error-detecting circuitry to counteract fault attacks increased information leakage and, as a result, simplified side-channel analysis. This can be seen as a "security paradox", where attempts to strengthen security against certain attacks can actually weaken the overall security of the system [5,6]. The objective of this work is to enhance the secure composition of physical attack countermeasures by identifying any potential negative consequences of specific countermeasures in other attack scenarios.

Various attack mechanisms can compromise cryptographic circuits, making them vulnerable. To prevent such attacks, several countermeasures have been proposed. Physical attacks, including algebraic fault attacks (AFA) and side channel attacks (SCA), are among the most prevalent types. AFA combines mathematical analysis and traditional fault attacks to recover the secret cryptographic key of a cipher. This technique derives equations from the algebraic descriptions of the cipher, fault-affected values, and a fault model. A SAT solver then processes the equations and the fault model to recover the secret key or limit the key space. SCA, on the other hand, captures side channel information such as power consumption or electromagnetic radiation to gather sensitive information about the cryptographic primitives. This is possible because the operation of cryptographic circuits can cause variations in power consumption or electromagnetic radiation that can be detected and analyzed. By capturing this information, an attacker can infer sensitive information about the cryptographic primitives, which can compromise the security of the system. Therefore, it is essential to implement proper countermeasures to prevent these attacks and ensure the security of cryptographic systems.

On the other hand, profit-driven globalization of integrated circuit design has resulted in many hardware-based attacks due to the outsourcing of fabrication to untrustworthy offshore foundries. As a result, new techniques are necessary to safeguard Integrated Circuits (ICs) against various attacks. One such technique is logic locking, which involves modifying the circuit and adding new locking key inputs. These locking keys must be set to the correct values for the circuit to function correctly. Unfortunately, since its inception, several attacks have successfully broken logic locking, i.e., unauthorized users can determine the locking key [7,8]. The most relevant among these is the SAT attack, which helps categorize logic locking approaches into "pre-SAT" and "post-SAT" methods. The latter have been specifically designed to withstand the SAT attack. In order to counter these attacks, many countermeasures have also been proposed [9–12].

In this article, we show that the logic locking schemes proposed for security against supply chain threats are a weak link in protecting cryptographic hardware. The cryptographic primitives are resistant against (classical) cryptanalysis mainly due to their algebraic properties, and we show that when an adversary can apply incorrect locking keys, logic locking can destroy this resistance. We demonstrate that differential fault analysis (DFA) on incorrectly unlocked cryptographic circuits is simpler than on regular circuits without logic locking. We call the resulting attack *locking-enabled DFA* or LEDFA. Moreover, the circuit can even become vulnerable to classical differential cryptanalysis with no physical faults injected for some locking keys. We refer to this attack as LEDA. Note that neither LEDFA nor LEDA aims to attack the logic locking scheme itself (in fact, we assume an adversary with legitimate access to the circuit and correct logic-locking key). The target of the attack is the cryptographic key used by the primitive, and the locking circuitry is used as a vehicle to facilitate this attack.

Our article has a two-fold objective. Firstly, we intend to provide a comprehensive exploration of the LEDA and LEDFA algorithms, with particular emphasis on potential vulnerabilities resulting from logic locking. We will look into the intricacies of both algorithms and highlight potential weaknesses that an attacker could exploit. We will also analyze successful attacks to demonstrate why our attacks are successful. In addition to [13], we have two new contributions. First, we propose a countermeasure against LED(F)A based on our findings and show empirical results to demonstrate its effectiveness. Secondly, we aim to also investigate the impact of logic locking on the side channel vulnerabilities of an AES module. We will conduct empirical assessments of power consumption traces to gain insights into the interaction between logic locking and the overall security profile of

cryptographic circuits. Our study aims to provide valuable insights into how logic locking can affect the security of cryptographic circuits and highlight potential areas of concern for designers and developers. Overall, our article contributes to the ongoing debate on the security of cryptographic circuits and provides insight into improving their robustness against attacks.

In order to demonstrate and study both LEDFA and LEDA, we developed a methodology which is summarized in Figure 1. Firstly, we generated implementations of cryptographic circuits logic locked using a representative selection of schemes from the three prominent logic locking families: pre-SAT, modification-based post-SAT, and restorationbased post-SAT logic locking schemes. To achieve this, we extended the existing circuit with the locking logic and additional inputs for the locking key *lk*, in addition to the regular inputs of the plaintext *p* and the cryptographic key *k*. Next, we ran DFA for the resulting locked circuit with the new *lk* inputs fixed to either the correct or random incorrect values. To accomplish this, we used an existing algebraic fault attack framework that was extended to support locking key inputs. The AFA framework generates a fault attack for a user-specified number of fault injections for a given fault model. Specifically, it constructs a Boolean formula in its conjunctive normal form (CNF) that can be solved for the cryptographic key by plugging in the fault-affected circuit responses.



Figure 1. Overview of the attack methodology (the first step is executed in our evaluation; in a real attack scenario, the circuit is locked already).

Our results show empirically that utilizing LEDFA with randomly selected incorrect locking keys is a more effective method than using DFA on a circuit that has no locking circuitry or has been correctly unlocked. Furthermore, LEDA, which does not require fault injections, has a success rate of 27% to 87% for lightweight ciphers, presenting a severe vulnerability since attackers can repeat the process using different incorrect locking keys until a successful attack. We conducted an extensive investigation into LED(F)A's success factors, including the impact of non-trivial locking circuitry in post-SAT schemes and the optimal choice of the locking key. As part of this investigation, we have demonstrated cases where LEDA can successfully attack full-scale AES. Overall, we show that logic locking is a

problematic technique for cryptographic hardware and goes against the principle of secure composition of countermeasures.

The remainder of the article is structured as follows. Section 2 provides essential background information on fault injection attacks, side channel analysis, and different logic locking schemes. In Section 3, we give a detailed motivational example. Section 4 summarises the individual attack steps and explains the chosen threat model. Section 5 reports on the experimental results. In Section 6, success factors are analyzed, including the choice of a specific locking key and the propagation of cryptographic key bits through locking subcircuits. Section 7 discusses a proposed countermeasure against LEDA. Finally, Section 8 concludes the article.

2. Background

Cryptographic primitives, the foundational building blocks ensuring secure data communication, have historically been the target of various sophisticated attacks. One such prevalent modality is the fault injection attack. Within this category, differential fault attacks have been particularly noteworthy due to their mechanism of leveraging fault injections to instigate faulty cipher executions. Intriguingly, these faulty executions only marginally deviate from fault-free operations. This variation facilitates differential cryptanalysis across a truncated number of cipher rounds, amplifying the efficacy of these attacks. The potency of differential fault attacks is underscored by their ability to fully derive keys for ciphers, including, but not limited to, AES-128 [14,15], LED [16], and PRESENT [17]. Often, a singular or minimal fault injection suffices for this derivation. The ensuing data created from such injections are commonly channelled into specialized [14] or generic equation-solving paradigms [18], with the latter often rooted in Boolean satisfiability solving.

Parallel to these, the hardware domain grapples with an array of attack vectors predominantly centred on supply chain security. These threats span malicious circuitry alterations during design or fabrication phases, commonly known as hardware Trojans [19], to more overt violations like piracy, cloning, counterfeiting, and overbuilding. The latter subset pertains to scenarios wherein circuits or their intellectual properties are unlawfully procured or originate from unauthorized entities [20]. These challenges are ubiquitously pertinent but assume higher significance in the context of cryptographic circuits.

Several countermeasures have been proposed to counter such threats. These include camouflaging—obscuring identifiable hardware architectures to thwart recognition [21]; logic locking—enabling circuit operability exclusively for authorized users possessing the requisite locking key [22]; watermarking—infusing the circuit with an indelible authorial signature [23]; fingerprinting—an augmented form of watermarking, it entails the inclusion of the purchaser's identity [24]; metering—a mechanism devised for the meticulous tracking of individual circuits [25].

2.1. Logic Locking

Introduced in 2008 as part of the EPIC ("Ending Piracy of Integrated Circuits") framework [26], random-logic locking (RLL) adds key gates connected to new locking key inputs at random locations within a given circuit. However, in 2012 [27], it was observed that an attacker with access to a functional (unlocked) circuit and a reverse-engineered netlist could make individual locking key bits observable at the circuit's outputs. To counter against this threat, strong logic locking (SLL), fault-analysis-based logic locking (FLL) [28], and look-up-table-based locking [29] were proposed. However, these techniques were compromised by the SAT attack [7], which used Boolean satisfiability (SAT) solving to compute "distinguishing input patterns" (DIPs) and thus reduce the key search space iteratively. The introduction of the SAT attack has led to the development of several new locking techniques such as SARLock [9], Anti-SAT [10], CASLock [30], TTLock [11], SFLL [12], and cyclic Locking [31].

In the recent literature, there exists a differentiation between pre-SAT and post-SAT logic locking techniques [32]. The former employs key gates inserted into the netlist,

varying only in their placement. The latter, on the other hand, utilizes point functions, which are Boolean functions that produce a value of 1 for only one point in the input space. This approach limits the extent of locking-triggered perturbation to the original circuit's functionality, thereby reducing the valuable information for an SAT attack. Another type of post-SAT scheme is the cyclic logic locking scheme, which introduces loops into the combinational circuitry, potentially oscillating under incorrect locking keys. However, these circuits violate standard design rules, and we consider them impractical for cryptographic applications. This study assesses a representative selection of both pre- and post-SAT techniques, including two varieties of RLL as the most basic pre-SAT schemes and two post-SAT schemes, namely Anti-SAT and SFLL. All of the schemes utilized are discussed in detail below.

2.1.1. Random-Logic Locking

As mentioned before, RLL randomly inserts locking key gates in a netlist. There are two different schemes considered for this work: random XOR/XNOR insertions and random AND/OR insertions. We refer to them as "RLL- \oplus " and "RLL- \wedge ", respectively, in the rest of the article. In RLL- \oplus , we select *n* wires in the circuit, where *n* is the size of the locking key, and then introduce *n* new XOR/XNOR gates. We disconnect each wire's driver from its sink and insert either an XOR or an XNOR gate. Whether we choose an XOR or an XNOR gate depends on the matched locking key bit. Similarly, in the case of RLL- \wedge , we randomly insert either an AND or an OR gate.

Let us consider Figure 2a representing an original circuit, which consists of three inputs and an output. It comprises five gates. On the other hand, Figure 2b shows the same circuit locked with RLL- \oplus using a 3-bit locking key. This variant has three additional XOR and XNOR key gates. At least one of the inputs of each key gate is driven by a wire from the original design, while the other input is driven by a locking key bit. The new locked circuit will only generate the correct output when activated using the correct locking key. For instance, consider Figure 2b again: if the locking key value of 110 is loaded into memory, the correct output for all input patterns (e.g., Y = 0 for input 000) is produced because all key gates will function as buffers. However, if an incorrect key value is used, certain key gates will function as inverters, resulting in errors in the circuit's output. For example, when the locking key 010 and the input 000 are applied, the key gate lk_1 will behave as an inverter, leading to an incorrect output Y = 1.



Figure 2. (**a**) A simple combinational circuit. (**b**) Circuit in (**a**) locked using random XOR/XNOR insertions.

2.1.2. Point Function-Based Logic Locking

Let us take a closer look at a generic point function-based logic locking scheme, shown in Figure 3a. These schemes utilize a *root* and an *tree input set* to embed the logic-locking block into a given circuit. The logic-locking block has as inputs the locking key *lk* and the tree input set, or *X*, which consists of a set of signals, either the primary inputs or intermediate wires. The size of this set is equal to the size of the locking key. The logiclocking block creates a point function Y, which is then connected to the randomly selected root within the original circuit. Our study investigates two schemes: Anti-SAT and stripped functionality logic locking (SFLL).



Figure 3. (a) Generic point function-based logic locking (b) Anti-SAT (c) SFLL.

Anti-SAT, the circuit block shown in Figure 3b, is relatively lightweight and consists of two functions, g and \overline{g} , that are connected to the same inputs, X, and locked with two n-bit keys, lk_1 and lk_2 . This results in a locking key size of 2n and a tree input set size of n. If the correct (lk_1, lk_2) is applied to unlock the circuit, the outputs of g and \overline{g} must be complementary. These outputs are then fed into an AND gate to produce a constant output of Y = 0, which does not affect the original circuit's functionality. However, suppose an incorrect locking key is applied. In that case, the complementary nature of g and \overline{g} is violated, and Y assumes the value 1 for some inputs, which leads to incorrect functionality of the original circuit.

The hardware implementation of *SFLL* intentionally removes parts of the design functionality through either gate insertions or replacements. As a result, the original design and the hardware implementation are not the same. A *restore unit* is added to recover the correct functionality, as shown in Figure 3c, which cancels the built-in error when the correct logic-locking key is applied. However, if an incorrect locking key is applied, it introduces additional errors in the design. With SFLL, designers can specify a permissible Hamming distance from the correct locking key. As a result, the restore unit recovers the original circuit behavior only when the locking key supplied by the circuit's user is within the specified Hamming distance from the correct one.

2.2. Fault Attacks

Cryptographic primitives can be weakened using fault-injection attacks, which cause physical disruptions during their operation. These attacks can be simple and can be achieved through bit manipulation or clock frequency manipulation. When a cryptographic operation produces a faulty result, it can be observed at the circuit's outputs, and the differences between the faulty and the fault-free results can be used to derive the secret information, such as the secret cryptographic key or the plaintext. Fault attacks involve physical fault injections on the circuit and the mathematical analysis of the observed data. The physical fault injections can be carried out in several ways, from glitching and underpowering to overheating [33], optical, and electromagnetic fault injections [34,35]. These attacks are invasive and active; for example, to shoot accurately at a specific location with a laser, it may be necessary to decapsulate the chip.

The mathematical analysis consists of manual derivation to either obtain the cryptographic key or reduce the keyspace to a size that allows a brute force search [14,16]. Statistical techniques like those discussed in [36,37] can also be employed. Algebraic fault attacks (AFA) use an algebraic formula to map an attack, as described in [18,38]. Improving the solving process can increase the scope of the suitable attacks. Algebraic cryptanalysis is computationally intensive but does not require specific external data. At the same time, fault attacks, on the other hand, are data-dependent and may demand many fault injections to recover the secret key. Combining the two attacks can benefit from both of their advantages. The main idea behind AFAs is to use the cipher description and some fault injection data to retrieve the sensitive information. In order to launch a successful attack, an appropriate fault model must first be selected. The cipher and the selected fault are then expressed in their proper algebraic form. Finally, a solver is used to solve the algebraic equations, and if necessary, the incorrect key candidates are eliminated.

2.3. Side Channel Attacks

Side channel attacks are a type of physical attack that aims to exploit vulnerabilities in cryptographic circuits to gain access to sensitive information. Unlike invasive attacks that involve tampering with the system, side-channel attacks monitor auxiliary outputs such as power consumption or electromagnetic emissions. These attacks target the correlation between the processed data and the secret cryptographic key.

Differential power analysis (DPA) is one of the most prominent types of side-channel attacks. DPA, introduced by [39], targets the correlation between the electrical power consumption of a cryptographic device and the operations it performs that depend on the secret cryptographic key. Statistical methods are used to analyze the power consumption, allowing the attacker to extract sensitive information.

The primary attack point of DPA is the dependence of power consumption on the operation performed by the cryptographic device. Multiple power consumption traces are measured for the fixed secret cryptographic key, and varying plaintexts are used. The resulting data are stored, including the measured power and the applied plaintext. With sufficient available traces, the first step of cryptanalysis is to specify a selection function that divides the measured traces into two sets based on an assumption of the secret cryptographic key. The average at each point in time is calculated and then compared for both sets. However, a large number of power traces are needed due to the presence of various hardware noises.

DPA has a critical advantage over brute force attacks because it allows the attacker to guess the key one byte at a time, significantly reducing the effort required and making DPA a highly effective tool for attackers seeking unauthorized access to sensitive information.

2.3.1. Test Vector Leakage Assessment (TVLA)

To evaluate the resilience of a cryptographic circuit against side channel threats, the TVLA method, as proposed by [40], stands as an efficient tool. TVLA aims to detect potential side channel vulnerabilities rather than explicitly extracting the secret cryptographic key. The procedure involves examining the circuit using a consistent and known set of inputs. Two distinct datasets are required to evaluate AES using TVLA to mitigate the risk of false-positive results. The initial dataset comprises power traces from 2 * n encryptions, all employing a consistent, predefined cryptographic key, k, but varied random plaintexts. In contrast, the second dataset employs the same cryptographic key, k, encrypting with a static input *n* times.

The objective is to determine the possibility of distinguishing between traces of fixed plaintext and the random counterparts, and is achieved through independent statistical evaluations with separate datasets. The first n/2 measurements from both datasets are compared for the initial test, while the subsequent test uses the latter n/2 measurements. A t-score outside the range of -4.5 to 4.5, specifically during the middle phase of the AES operation, indicates a failed test. In the context of the 4.5 threshold, [41] states it offers "a confidence of >0.99999 to reject the null hypothesis". This t-score corresponds to the t-statistic in Welch's *t*-test, where a significant deviation from zero suggests a discernible difference between the two sets. Therefore, to meet the requirements of the TVLA, a small absolute value of the t-statistic is desired. Notably, a system is more vulnerable to a side channel attack if it fails the TVLA. However, passing the TVLA does not conclusively prove that the circuit can resist side-channel attacks. Even with a successful test outcome, it is essential to corroborate the findings through additional evaluation methodologies.

2.3.2. *t*-Test

After collecting the traces, Welch's *t*-test is conducted on the datasets—the *t*-test tests whether two sets exhibit the same mean. Thus, a test failure would indicate that statistically significant differences in mean are present between the datasets. In the context of TVLA, the rejection of the null hypothesis—that the two datasets possess identical means—implies security vulnerabilities within the device under inspection [42]. Contrarily, the acceptance of the null hypothesis suggests an absence of discernible security vulnerabilities. However, it is crucial to note that the absence of evidence does not guarantee the device's intrinsic security.

For an implementation of TVLA, it is specified that two independent tests be executed on mutually exclusive datasets. The first test is done with the initial half of the dataset, while the second test evaluates the latter half. The t-statistic is meticulously calculated for both tests. Let us denote X_1 and X_2 as the means of the random and fixed datasets at any specified point within the set of sample points. Similarly, let S_1 and S_2 represent the sample standard deviations of the random and fixed datasets, while N_1 is the number of traces used to test the random dataset and N_2 is for the fixed dataset. The t-statistic is thus expressed by:

$$t = \frac{X_1 - X_2}{\sqrt{\frac{S_1^2}{N_1} + \frac{S_2^2}{N_2}}} \tag{1}$$

After the t-statistic computation across all sample points, whether these values reside within the interval [-4.5, 4.5] is determined. Any deviation of the t-statistic from this specified interval implies the rejection of the null hypothesis, indicating potential security vulnerabilities. The choice of this interval, precisely [-4.5, 4.5], aligns with a confidence level of 99.999%, thereby minimizing the probability that observed disparities between datasets arise merely from stochastic variances [40].

3. Motivational Example and Attack Assumptions (Rephrased)

In this segment, our objective is to provide a clear explanation of the concept of lockingenabled attacks. To achieve this, we will present an example of LEDFA that is based on a basic but rather unrealistic scenario, which will help in understanding the idea in a simple way. Following that, we will introduce our actual adversary and threat model in the next section, providing a more practical perspective. We implemented a gate-level version of the PRESENT SBox with a Boolean function $S : \mathbb{B}^4 \to \mathbb{B}^4$, and locked it with a 4-bit locking key using RLL- \wedge to illustrate the concept. This changed the Boolean function of the SBox to $S_l : \mathbb{B}^8 \to \mathbb{B}^4$ and led to the addition of four new inputs, lk_1, \ldots, lk_4 . Only one lk among the 16 potential locking keys is correct; in that instance, for all $i \in \mathbb{B}^4$, $S_l(i, lk) = S(i)$. The adversary can apply any other $lk' \neq lk$, which will change the SBox's function, or the correct lk = 0110 to use it normally.

Now, we will show how logic locking can make the SBox and the cipher as a whole more susceptible to differential cryptanalysis. The *difference pairs* $(\Delta i, \Delta S(i))$ for the SBox will be computed for this purpose. These are pairs of inputs (i', i'') that satisfy $i' \oplus i'' = \Delta i$ and $S(i') \oplus S(i'') = \Delta S(i)$ [43]. Table 1A displays the resulting *difference distribution table* (DDT) for the standard SBox without any locking circuitry (or, equivalently, a locked circuit unlocked with the appropriate locking key). For instance, entry 2 in row 7 and column 2 shows that $S(i') \oplus S(i'') = 2$ is equal to the number of input pairs (i', i'') with $i' \oplus i'' = 7$. Aside from the simple case $\Delta i = \Delta S(i) = 0$, this is also the largest number in the DDT. Greater DDT entries suggest that the SBox is more vulnerable to differential cryptanalysis. Since there is no such thing as a mathematically impossible "ideal" SBox, all differential values would be 1, suggesting maximum resistance.

Table 1. Function table of the standard (**A**) and the incorrectly unlocked (**B**) PRESENT SBox and difference distribution tables for the standard (**C**) and the incorrectly unlocked (**D**) PRESENT SBox. Here, Δi and $\Delta S(i)$ stand for number of input pairs (i', i'') that satisfy $i' \oplus i'' = \Delta i$ and $S(i') \oplus S(i'') = \Delta S(i)$, respectively.

	Input <i>i</i>	0	1	2	3	4	5	6	7	8	9	А	В	С	D	Е	F
0	utput S(i)	С	5	6	В	9	0	А	D	3	Е	F	8	4	7	1	2
							(A)									
	Input <i>i</i>	0	1	2	3	4	5	6	7	8	9	А	В	С	D	Е	F
Out	put $S_l(i, lk')$	7	7	5	5	7	7	7	7	5	5	7	5	7	5	5	5
							(B)									
					Out	out T	Diffe	rence	$\Delta S($	<i>i</i>)							
		0	1	2	3	4	5	6	7	8	9	А	В	С	D	Е	F
	0	16	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	1	0	0	0	4	0	0	0	4	0	4	0	0	0	4	0	0
	2	0	0	0	2	0	4	2	0	0	0	2	0	2	2	2	0
	3	0	2	0	2	2	0	4	2	0	0	2	2	0	0	0	0
Δi	4	0	0	0	0	0	4	2	2	0	2	2	0	2	0	2	0
ICe	5	0	2	0	0	2	0	0	0	0	2	2	2	4	2	0	0
ren	6	0	0	2	0	0	0	2	0	2	0	0	4	2	0	0	4
ffe	7	0	4	2	0	0	0	2	0	2	0	0	0	2	0	0	4
Dij	8	0	0	0	2	0	0	0	2	0	2	0	4	0	2	0	4
ut]	9	0	0	2	0	4	0	2	0	2	0	0	0	2	0	4	0
du	А	0	0	2	2	0	4	0	0	2	0	2	0	0	2	2	0
Ξ	В	0	2	0	0	2	0	0	0	4	2	2	2	0	2	0	0
	C	0	0	2	0	0	4	0	2	2	2	2	0	0	0	2	0
	D	0	2	4	2	2	0	0	2	0	0	2	2	0	0	0	0
	E	0	0	2	2	0	0	2	2	2	2	0	0	2	2	0	0
	F	0	4	0	0	4	0	0	0	0	0	0	0	0	0	4	4
					_		(C)		(
		0		•	Out	put L	Diffe	rence	$\Delta S($	<i>i</i>)	0			6	P		-
		0	1	2	3	4	5	6	7	8	9	A	В	C	D	E	F
	0	16	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	1	12	0	4	0	0	0	0	0	0	0	0	0	0	0	0	0
	2	8	0	8	0	0	0	0	0	0	0	0	0	0	0	0	0
•••	3	8	0	8	0	0	0	0	0	0	0	0	0	0	0	0	0
Ā	4	8	0	8	0	0	0	0	0	0	0	0	0	0	0	0	0
JCe	5	8	0	8	0	0	0	0	0	0	0	0	0	0	0	0	0
rer	6	12	0	4	0	0	0	0	0	0	0	0	0	0	0	0	0
ffe	7	8	0	8	0	0	0	0	0	0	0	0	0	0	0	0	0
Di	8	4	0	12	0	0	0	0	0	0	0	0	0	0	0	0	0
ut	9	4	0	12	0	0	0	0	0	0	0	0	0	0	0	0	0
du	A	ð	U	ð	0	0	0	0	0	0	0	U	0	0	0	0	0
Π	D C	ð o	0	ð o	0	0	0	0	0	0	0	0	0	0	0	0	0
		ð	0	ð o	0	0	0	0	0	0	0	0	0	0	0	0	0
	D F	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
	E F	4 1	0	12 12	0	0	0	0	0	0	0	0	0	0	0	0	0
	Г	4	U	12	0	U	U	0	U	U	0	U	U	U	U	0	0
							(D)									

We unlocked the PRESENT SBox with all possible locking key combinations, and then we exhaustively computed the differential distribution table (DDT) for every resulting function, $S_l(\cdot, lk')$. Because the resulting function $S_l(i, lk')$ for lk' = 0000 had the weakest DDT against differential cryptanalysis among all the resulting functions, we decided to use lk' = 0000 for our motivational example. The corresponding DDT, shown in Table 1D, is susceptible to differential cryptanalysis because it contains multiple entries with a value of 12. Furthermore, there exist several input differences for which the resulting "SBox" is not a bijective mapping, as evidenced by the fact that the output difference is 0.

Next, we attempted DFA on an implementation of PRESENT where four out of 16 SBoxes in round 28 were replaced by the locked version $S_l(\cdot, lk')$ with lk' determined by the DDT analysis above. This is shown in Figure 4 (using three faults according to the nibble fault model explained in detail further below). Four copies of the modified PRESENT (one fault-free and three fault-affected circuits) were used to build a differential model, which we then converted into conjunctive normal form and solved using the SAT solver CryptoMiniSAT.



Figure 4. Faulty PRESENT encryption with 4 incorrectly unlocked SBoxes (in red) in round 29 (pLayer denoted as pL).

The outcomes of this experiment, along with another using the same SBox and LED in place of PRESENT, are displayed in Table 2. The percentage of attack attempts that yielded the correct cryptographic key is indicated by the "success rate". Once the remaining attack attempts had been solved for 18,000 s, they timed out. It is observed that the complexity of attacks was significantly lower for implementations with incorrectly unlocked SBoxes. With LEDFA, attacks that were previously unfeasible for DFA become possible, and the run time is drastically decreased. This is not surprising, as the DDT analysis shows that the cryptographic strength of the SBoxes has been significantly diminished.

Table 2. Performance of DFA and LEDFA on motivational examples (50 repetitions).

	D	FA	LEDFA	A
Cipher	Success Rate [%]	Average Time [s]	Success Rate [%]	Average Time [s]
PRESENT (3 faults)	0	_	100	5587.2
LED (1 fault)	100	793.2	100	75.9

The example in this section shows how LEDFA can be used, but more discussion is needed to fully understand its viability. It is a realistic assumption that adversaries can gain access to the circuit by using both the correct and the incorrect locking keys. The right locking key (lk) must be applied by legitimate users in order to confirm their authorization. As a result, an externally accessible interface is required to provide locking keys. Given that anyone purchasing the circuit can easily obtain the correct locking key, it is also conceivable that adversaries are aware of it. However, the cryptographic key k is not accessible from the outside as it is kept in a protected memory inside the circuit. Recovering this cryptographic key k is the primary goal of LEDFA. Importantly, the adversary does not need to exert any further effort in order to apply an incorrect locking key. Simply said, they can apply various logical values to the circuit's standard inputs.

In the example, it is assumed that the four locking key gates of RLL- \land have affected the four SBoxes of the circuit in a particular way. If the locking scheme applied to the entire PRESENT circuit includes four key gates in each of these four SBoxes at precisely the same positions as determined by the DDT analysis, then such an effect of logic locking could occur. This assumption, however, is improbable because the key gates are inserted during the manufacturing process and may be positioned at random or in response to the structural characteristics of the circuit. They can be placed in any of the other blocks or within the SBoxes. It is crucial to remember that adversaries are limited to using the provided locked circuit and cannot move or add new key gates. We therefore adhere to the white-box assumption, which states that although the adversary is unaware of the values of the cryptographic secret key k, they are aware of the gate-level implementation of the cryptographic circuit, including the circuitry related to locking. It is also assumed that there are no specific countermeasures included to thwart or impede fault attacks. Future research, though, can address this in more detail.

The next section will include a more thorough description of the LEDFA attack steps along with a more formal summary of the threat model.

4. Attack Overview and Threat Model (Rephrased)

A cryptographic function enc is assumed to have a logic-locked implementation enc_l. Three inputs are provided to enc_l: plaintext p, cryptographic key k, and locking key lk. The function computes a ciphertext $c = \text{enc}_l(p, k, lk)$. An adversary can choose a p to apply to the circuit's inputs and then record the resulting ciphertext in accordance with common assumptions about differential fault attacks. The adversary's goal is to recover the value of the cryptographic key k, which is safely stored within the circuit. Adversaries also have the ability to apply various values to lk as an input. For any given specification (enc_l(p, k, lk) = enc(p, k) $\forall p, k$), there is only one correct locking key lk fitting the description. Incorrect behavior will arise from applying an incorrect $lk' \neq lk$; however, tracking or detection is absent, i.e., the adversary can use the circuit with the wrong locking keys, and it will not have any negative effects, like output randomization or blinding.

A physical disturbance represented by a fault f is injected into the circuit while it is being used for the differential fault analysis (DFA). Generally speaking, an attack can include several fault injections, i.e., repeatedly utilizing the cryptographic circuit for encryption with faults $f_1, f_2, ...$ The behavior of the circuit under f_i fault is given by $\operatorname{enc}_l^{f_i}(p, k, lk)$. The adversary might not be completely aware of or in control of the fault injection's effect in most situations. Thus, it is unknown what intermediate values are impacted. With the same inputs, p and lk, the adversary records the ciphertext Cgenerated by the fault-free circuit and the faulty ciphertexts $C_1, C_2, ...$ generated by the same circuit, upon injection of the corresponding faults, $f_1, f_2, ...$ The cryptographic key is then recovered by solving the equation

$$(\operatorname{enc}_{l}(p,k,lk) = C) \wedge (\operatorname{enc}_{l}^{f_{1}}(p,k,lk) = C_{1})$$
$$\wedge (\operatorname{enc}_{l}^{f_{2}}(p,k,lk) = C_{2}) \wedge \dots$$
(2)

after *k*. To build the equation, we assume that the adversary is aware of the gate-level implementation of the circuit, including the locking circuitry. Reverse engineering [44] can

be used to determine this information, even though it is not always available in practice; in DFA literature, it is typically assumed that the attacked function is known.

The presence of input lk sets our proposed attack apart from other traditional fault attack techniques. As was shown in the previous section, the function can be changed to a cryptographically weaker version, $enc_l(\cdot, \cdot, lk') \neq enc$, if an incorrect locking key $lk' \neq lk$ is used. If the incorrect locking key is used to mount the fault attack, Equation (2) may become easier to solve.

To perform LEDA, we use fault-free encryptions; we generate multiple ciphertexts with distinct locking keys: $C_1^l = \text{enc}_l(p, k, lk_1), C_2^l = \text{enc}_l(p, k, lk_2), \ldots$ Take note that lk_1, lk_2, \ldots need to be pairwise different; the correct locking key can be one of them, but it is not required. Solving the equation

$$(\operatorname{enc}_{l}(p,k,lk_{1})=C_{1}^{l})\wedge(\operatorname{enc}_{l}(p,k,lk_{2})=C_{2}^{l})\wedge\dots$$
(3)

after the cryptographic key, *k* becomes computationally tractable for at least some cases. In this case, using the wrong locking key has a function that is comparable to fault injections in DFA. There are, nevertheless, a few significant differences. Unlike many injection techniques, LEDA does not suffer from inadequate spatial or temporal resolution and does not require fault-injection equipment. Since there are no peculiar circuit conditions as a result of the physical disturbances, it is rather challenging to detect using the built-in sensors. On the downside, since the locking circuitry does not affect every location in the circuit, LEDA cannot cause arbitrary faults (from the adversary's perspective). Nonetheless, we demonstrate that the areas impacted by the locking key are frequently adequate for recovering the secret cryptographic key. Furthermore, it should be noted that Equation (3) may not be computationally tractable for arbitrarily large values of lk_1, lk_2, \ldots ; in fact, we found a less than 50% success rate in all cases that were examined. This is not a major drawback for an adversary who can simply keep trying other locking key combinations until one recovers *k*.

Remember that in order to illustrate LEDA and LEDFA, Figure 1 shows how the attack is set up in general; recall specifically that logic locking is applied to a cryptographic circuit is the first step. In an actual attack scenario, it is assumed that the locked circuit already exists. The remaining steps of the procedure show how to use the AFA framework, AutoFault, which is described in [38], to launch an algebraic fault attack against the locked circuit. The ensuing subsections contain more specific information about each step.

4.1. Time-Frame Expansion

Two methods can be used for the hardware implementation of round-based ciphers: one is to use a sequential circuit where the logic for one or more rounds generates intermediate outputs that are stored in registers, or one is to use a combinational (unrolled) circuit that contains logic for each round. Propositional models, which are needed for the SAT solvers at the core of the AFA framework used in this work, can be obtained by applying the Tseitin transform [45] to combinational circuits in order to convert them into a Boolean formula. Unrolling or time frame expansion must be used to create a functionally equivalent combinational circuit for a sequential circuit. In essence, this procedure is a one-cycle implementation of the circuit's original description. It is important to remember that test pattern generation for sequential circuits or sequential equivalency checking commonly employs the time frame expansion technique.

4.2. Logic Locking

We used a publicly available synthesis program named Netlist Encryption and Obfuscation Suite (NEOS) [46] to generate the locked circuits. Random-logic locking variants comprise seven of the available schemes. For this work, we consider two of them, XOR/XNOR and AND/OR insertions at random. Additionally, three logic locking schemes based on point functions are supported by the tool. As an example of these, we take Anti-SAT and SFLL. Furthermore, as noted in Section 2.1, we believe that the four cyclic locking schemes that NEOS supports are not suitable for use in cryptographic implementations. Universal logic locking, the final supported scheme, maps the circuit to a structure resembling an FPGA, with the configuration bits acting as the locking key once they are set after manufacturing. The adversary could easily prevail in our threat model, where users have the ability to apply arbitrary locking keys, by setting up the FPGA to propagate the cryptographic key to all of its outputs.

NEOS creates a locked version of the given (in our case, cryptographic) circuit after receiving a gate-level netlist of the circuit. We had to re-synthesize our circuits before inserting them into NEOS since it can only process specific gates (Figure 1). The user can specify the number of locking key bits in addition to the locking scheme to be used; in our experiments, we generated circuits locked with 32, 64, 80, and 128 bits.

In our experiments, logic locking is applied *after* the circuit is unrolled using time frame expansion. As a result, the matching locked cryptographic circuit is combinational, meaning that key gates can be positioned anywhere in the circuit and that it is completely unrolled and functional in a single clock cycle. On the other hand, in a sequential implementation, the key gates would be positioned inside the combinational core and would, therefore, appear in each frame. By applying the appropriate locking key bits to the key gates that are not required in a given clock cycle, an attacker can activate or deactivate specific key gates in each clock cycle, assuming that locking key inputs are regular circuit inputs. As a result, there would always be an equality between the sequential case and the combinational cases. This was done specifically in order to streamline the tool flow.

4.3. CNF Generation

Equations (2) and (3), which are represented by propositional (Boolean) formulae in conjunctive normal form (CNF), serve as the foundation for the LEDA and LEDFA attacks. A Boolean function that could be mapped to CNF is implemented by a combinational circuit with n input and m output made up of logic gates; however, this mapping is not very effective. For every logic gate in the circuit, the Tseitin transformation produces a *characteristic function*. For consistent assignments to the inputs and outputs of the corresponding gate, the characteristic function is 1. For instance, the characteristic function of an AND gate with inputs a, b and output c is

$$\chi_{AND} = (c \equiv (a \land b)) = (\overline{c} \lor a) \land (\overline{c} \lor b) \land (\overline{a} \lor b \lor c).$$

The conjunction of the characteristic functions of each gate in a circuit is the circuit's characteristic function. It is satisfiability equivalent to the circuit's Boolean function; how-ever, it has additional variables for internal signal lines.

The characteristic function CNF can have unit clauses (s_i) or $(\overline{s_i})$ added to it, respectively, to force a chosen signal line s_i to 1 or 0. A SAT solver for such a formula can report that the formula is unsatisfiable or yields a consistent assignment to the circuit's inputs and other variables where s_i has the desired value for the given assignment. The output restrictions "= C_j " in Equations (2) and (3) are implemented using this technique, where C_j is a constant vector made up of ciphertext bits that are physically observed on the outputs of the circuit under. The Boolean variables or literals corresponding to the output bits are found and added to the CNF as unit clauses, which are negated in the case that the observed bit was 0 and non-negated in the case that it was 1.

4.4. Attack Construction and SAT Solving

The formula for LEDA (Equation (3)) or the fault-assisted LEDFA (Equation (2)) is built using the AFA framework AutoFault [38]. The (time-frame-expanded) circuit for LEDFA has *k* copies in the attack formula, along with *k* fault injections. To build CNF parts for terms such as $\operatorname{enc}_{l}^{f_{i}}((p, k, lk) = C_{i})$, more details regarding injected faults are required. Included in this is the fault model (e.g., stuck at fault, bit flip, delay-based glitch), the round in which the fault is injected, the position of the fault within the round, and the number of faults injected concurrently. We employ a popular *random byte fault* model for DFA in this work. An arbitrary subset of bits within that byte can be flipped; the attacker is oblivious to which bits have been flipped. It is assumed that the injected fault affects one byte of the cipher state.

An SAT solver was given the attack formulae that were created with the AFA framework. We tested a number of SAT solvers, and since CryptoMiniSat [47] produced the fastest solving times, we used it for all of our experiments. The attack formula typically does not have a unique solution because of the random byte or nibble faults used, and it may produce a key candidate that is not the correct cryptographic key. In order to confirm these cryptographic key candidates, an encryption with a known plaintext/ciphertext pair must be performed. A conflict clause is added to the CNF to remove the candidate from the solution space and rerun the attack if it turns out not to be the correct cryptographic key.

Equation (3) describes the method used for the fault-free LEDA in the same way. Nevertheless, rather than modeling fault injections as seen in Figure 5, the formula consists of copies of the fault-free circuits with their locking key inputs and outputs fixed to different lk_i and C_i^l , respectively. A considerable portion of LEDA cases were unable to extract the cryptographic key in the allotted time. This makes sense since improper unlocking may affect the underlying cryptographic construction's ability to withstand attacks, but it might not always. Again, keep in mind that lk_i is the circuit's primary input. If an attacker's initial attack times out after 18,000 s, they can simply launch another one at no additional expense.



Figure 5. Attack construction for locking enabled differential analysis (LEDA).

5. Experimental Results

In this study, we carried out three sets of experiments. Our first results correspond to fault attacks (LEDFA) on locked cryptographic circuits. We used randomly generated faults to attack four ciphers: PRESENT, LED-64, and AES-64, as well as the full-scale AES. In our second experiment, we performed differential analysis on locked ciphers (LEDA) using randomly generated incorrect locking keys. To ensure accuracy, we recorded the average solve times taken by each attack on AMD Ryzen 9 3950X machines utilising 4 cores, 64 GB memory, and CentOS 8 as the operating system. For repeatability, we used pseudorandom plaintext, a cryptographic key, a locking key, and fault(s) generated from a recorded random seed. It is essential to note that the cryptographic key was only used for calculating fault-free and faulty ciphertexts and not during the attack itself. We set a timeout of 18,000 s and used a suitable number of faults for each cipher to ensure the attack did not exceed this timeout. Finally, for our third experiment, we conducted a test vector leakage assessment of AES to conclude our testing.

5.1. Locking-Enabled Differential Fault Analysis (LEDFA)

In this section, we present the results of the fault attacks performed on the selected ciphers: PRESENT, LED, and AES. We applied four different logic locking schemes— $RLL-\oplus$, RLL- \wedge , Anti-SAT, and SFLL—to secure the circuits. Table 3 provides an overview

of the cipher parameters and the size of the SAT formulas before and after applying a 128-bit locking key to the circuit. As observed, the number of clauses in the formulas is 3–4 times greater than the number of gates. This is common for circuits with numerous two-input gates. The addition of logic locking has led to a slightly reduced number of clauses, as the CNF generation procedure has redundancy elimination built-in.

Table 4 presents the results of the experiment. It is clear from the table that LEDFA outperforms the conventional DFA technique (row "None"). Among the LEDFA variants, RLL is the fastest for PRESENT and LED lightweight ciphers, while Anti-SAT and SFLL are significantly slower. However, for the larger full-scale AES, the trend is reversed. It is worth noting that interpreting the run time of a SAT solver is tricky due to the NP-completeness of the problem and various built-in speed-up heuristics of modern solvers. We believe that the run time differences are due to two opposing trends. On the one hand, the RLL techniques employ multiple key gates, and the extent of perturbation caused by an incorrect unlocking is related to the ratio of locking key gates to "regular" gates within the circuit. The likelihood that RLL will disrupt or weaken the algebraic properties of the cipher to such a degree that the resulting formula is easier (faster) to solve is higher for smaller circuits because an incorrect key bit is more likely to "hit" a critical part of the circuit.

Table 3. Numbers of gates **#G** and clauses **#C** before and after locking using a 128-bit locking key *B* and *#R* indicate the block size in bits and the number of rounds of the respective cipher.

Analyzed Cinhar (D #D)	No Locking		RLL-⊕		RLL-		Anti-SAT		SFLL	
Analyzed Cipher (b, #K)	#G	#C	#G	#C	#G	#C	#G	#C	#G	#C
AES (128, 10)	190,807	615,450	190,935	590,983	190,935	590,855	191,064	591,368	175,570	591,753
LED (64, 32)	26,104	94,786	26,232	84,643	26,232	84,515	26,361	85,028	23,426	85,413
PRESENT (64, 31)	34,087	110,056	34,215	102,264	34,215	102,136	34,344	102,649	31,162	103,034

Table 4. Average LEDFA run times in seconds for PRESENT (100 repetitions, faults injected after round 29), LED-64 (100 repetitions, faults injected after round 30), "AES-64" (50 repetitions, faults injected into round 8), and full-scale AES (30 repetitions, faults injected into round 8).

Locking	Locking	PRI	ESENT	I	LED	"AES-64"	AES-128
Method	Key Size	10 Faults	5 Faults	2 Faults	1 Fault	1 Fault	2 Faults
None	-	31.550	1378.940	17.136	793.198	9564.60	15,684.00
	32	2.028	93.757	12.818	596.197	1513.96	_
	64	1.159	95.821	9.557	136.984	2219.56	10,927.30
KLL-⊕	80	1.014	1.827	15.066	691.485	2998.38	12,765.20
	128	0.790	1.028	11.602	312.700	1932.39	11,876.21
	32	1.722	56.892	9.062	368.305	12,271.90	-
	64	1.010	22.384	15.361	222.264	7505.51	11,876.20
KLL-/\	80	0.879	12.252	11.351	260.450	5041.83	13,732.37
	128	0.751	5.936	29.261	218.757	4489.41	10,896.32
	32	6.881	57.050	8.344	232.000	3149.90	-
Anti CAT	64	6.913	550.466	8.974	288.200	2312.07	2537.10
Anti-SAI	80	11.910	623.625	9.336	381.490	3510.11	3501.56
	128	6.473	1369.525	8.978	196.687	2717.83	3487.98
	32	9.566	83.984	10.769	314.879	5423.79	_
CEI I	64	10.578	264.300	11.246	344.279	3453.09	10,768.37
JELL	80	14.876	328.134	12.454	389.564	3097.10	11,287.75
	128	15.473	211.970	12.568	225.674	2344.09	9978.19

It is important to note that post-SAT locking methods can introduce an additional attack path. This is because a subcircuit generates the signals used for locking, and bits of the cryptographic key can be input into this subcircuit. The impact of this mechanism is evaluated in the context of LEDA attacks in Section 6.2. For easy-to-solve instances, propagating the cryptographic key bits through the regular cipher logic (weakened by

incorrect unlocking) is sufficiently fast, and the alternative propagation mechanism does not play a significant role. However, for full-scale AES that is on the limit of the SAT solver's capabilities, the presence of a further propagation path through a subcircuit that was not designed to be cryptographically strong can lead to a measurable speed-up. The findings for LEDA in Section 6.2 are consistent with the fact that Anti-SAT facilitates LEDFA more than SFLL. One reason for this is the higher density of cryptographic key bits among the subcircuit's inputs.

All the attack runs in Table 4 were carried out within the given timeout of 18,000 s, resulting in a 100% success rate. We tried to identify the boundary condition when LEDFA starts to fail. In order to determine this, we conducted an experiment with PRESENT, using only three faults and increased the timeout to 25,000 s to determine the failure point of LEDFA. The results showed that LEDFA had a success rate of 0.55, with an average solve time of 18,827.87 s out of 20 repetitions when PRESENT was locked using Anti-SAT with an 80-bit key. On the contrary, DFA on PRESENT with three faults had timed out without exception, resulting in a 0% success rate. Therefore, it is evident that while the LEDFA attack enables key recovery, the DFA attack does not.

5.2. Locking-Enabled Differential Analysis (LEDA)

In this section, we will discuss the findings of the LEDA attack on three lightweight ciphers, namely LED, PRESENT, and small-scale "AES-64" (as shown in Figure 5). Table 5 presents the success rates and average solving times in seconds for each cipher. We locked each circuit using a 64-bit locking key using multiple schemes, including RLL- \oplus , RLL- \wedge , Anti-SAT, and SFLL. For LED and AES 4-4-4, we performed two encryption operations using the same plaintext and cryptographic key. For PRESENT, we performed ten encryption operations using the same plaintext and cryptographic key. Algebraic equations were then generated for each case according to Equation (3), which were converted into their corresponding Boolean formulae using the Tseitin transform and given to a SAT solver.

	#Incorrect	RLL- ⊕		R	LL-A	An	SFLL	
Cipher	Locking Keys	Success Rate	Solve Time [s]	Success Rate	Solve Time [s]	Success Rate	Solve Time [s]	Success Rate
PRESENT	10	0.567	2143.79	0.767	2348.12	0.867	3596.20	0
LED	2	0.367	698.05	0.267	869.34	0.533	1018.66	0
"AES-64"	2	0.433	9864.20	0.700	10,923.78	0.733	13,923.30	0

Table 5. Success Rates and average LEDA run times in seconds for successful attacks on PRESENT, LED-64, and "AES-64", locking key size of 64-bits, 30 repetitions.

Based on the results depicted in the table, we can see that the success rate of attacks on the ciphers under consideration ranges from 27% to 87%, implying that an attacker may need to make multiple attempts before succeeding. The multiple attempts are clearly feasible, especially given that the locking keys are simply applied to the inputs, which is easier than performing a fault injection in DFA. It is important to note that LEDA was only effective for lightweight ciphers within the framework used and consistently timed out for full-scale AES with random locking keys on fixed locations. However, in Section 6.2, when exploring alternative locking circuitry constructions, we discovered that LEDA could break an AES-128 implementation locked by Anti-SAT and even by SFLL. It is evident that this attack is at the limit of what a modern SAT solver can do. Ultimately, LEDA can effectively break lightweight ciphers that are deemed secure without requiring any physical manipulations or measurements beyond regular access to the circuit's inputs and outputs.

5.3. Test Vector Leakage Assessment of Logic Locked AES

In this work, we aim to examine the effect of logic locking on the vulnerability of cryptographic circuits against side-channel attacks. We chose to use the hardware imple-

mentation of AES and apply two different logic locking schemes, Anti-SAT and SFLL, to lock it. We then conducted a comprehensive power analysis of the circuit's power consumption by performing multiple encryptions with both the AES circuits locked using a 128-bit Anti-SAT locking key and the AES circuits locked with a 128-bit SFLL locking key. By doing so, we can compare the leakage with that of an AES circuit with no logic locking applied.

To evaluate the impact of leakage, we used the general fixed vs. random data method, as mentioned in [40]. This method allows us to determine the differences in power consumption when applying different data to the underlying design and whether these differences can potentially be exploited for side-channel attacks. In this case, we collected measurements for two different datasets: one with fixed input and the other with randomized input. We then carried out a statistical test on these datasets to determine whether they showed a statistically significant difference in mean or variance. If the test was successful, this would indicate that the design leaks information about the processed data.

The following sections will discuss the measurement setup, collection of traces, and the results. We will also present the results of the TVLA and explain how the leakage was determined. Overall, this study aims to demonstrate the increased vulnerability of cryptographic circuits to side-channel attacks in the presence of logic locking.

5.3.1. Measurement Setup

We implemented logic-locked cryptographic circuits on the Sakura-G FPGA board, equipped with two Xilinx Spartan-6 FPGAs. The primary FPGA, Spartan-6 LX-75, handled cryptographic computations, while the secondary FPGA, Spartan-6 LX-9, acted as a controller, providing a USB interface for external communication. To ensure precise measurement of the primary FPGA's power consumption, the board provides an amplifier and a dedicated connection point for oscilloscope interfacing. For data collection, we used Teledyne LeCroy's HDO6104A-MS oscilloscope, which provides a bandwidth of up to 1 GHz. To adhere to the recommended sampling rate of five times the oscilloscope's bandwidth, we adopted a 5 GHz sampling rate, as suggested by [40], for accurate results.

5.3.2. Collection of Traces

Let us consider the data collection for AES logic locked with a 128-bit Anti-SAT locking key. The encryption function then depends on three inputs: the plaintext p, the locking key lk, and the secret cryptographic key k and is represented as $AES_{AS}(p, lk, k)$. The application of TVLA requires the collection of at least n traces for each of the two datasets. The random dataset contains the randomized inputs, while the fixed dataset contains the fixed inputs. For both experiments, the secret cryptographic key on the device is set to the same fixed value k for all encryptions. For the fixed dataset, the inputs plaintext and locking key are set to the values p_{fixed} and lk_{fixed} for each encryption. The fixed dataset only contains measurements for $AES_{AS}(p_{fixed}, lk_{fixed}, k)$.

Random plaintexts and random locking keys are used for the randomized dataset. The locking keys are randomly generated 128-bit values. The next plaintext p_i is generated by encrypting the last plaintext $p_{(i-1)}$ using a separate cryptographic key k_{gen} . The encryption to generate the next plaintext is done using a software implementation. Then, a power measurement is recorded for $AES_{AS}(p_i, lk_i, k)$. The values for p_{fixed} , lk_{fixed} , k, and k_{gen} are randomly generated, while the initial plaintext p_0 is set to all zeros. A total of n = 50,000 power traces were collected for each dataset. A warm-up phase for the circuit was conducted before starting the measurements. Ten thousand encryptions for all randomly generated values were executed. After the warm-up phase, the measurements were started. The collection was interspersed, i.e., for each iteration, it was randomly decided whether the subsequent encryption should be for the fixed dataset or the random dataset. The process is then run until both sets contain at least 50,000 traces.

5.3.3. Results

In order to analyze the data collected, we performed TVLA and two *t*-tests for each experiment. To perform the *t*-tests, we divided the collected datasets into two groups, with the first half being used for the first test and the second half being used for the second test. After dividing the data, we calculated the mean and standard deviation for each point in time and used these values to determine the t-statistic for Welch's *t*-test. The resulting test data is displayed in Figure 6, with subfigures (a), (b), and (c) representing the results for AES, AES with a 128-bit Anti-SAT key, and AES with a 128-bit SFLL key, respectively. The encryption process for all the graphs can be observed between the two dotted red lines.





The circuits locked using Anti-SAT and SFLL were chosen based on the number of cryptographic key bits fed into the locking subcircuit. The Anti-SAT circuit had 64 bits, while the SFLL circuit had only 29 bits. For both locked circuits, we observed significantly higher leakage during the encryption process compared to the unlocked AES implementation. However, the t-values were slightly lower for the SFLL circuit. This can be attributed to the lower number of cryptographic key bits processed by the locking subcircuit, i.e., the 29 bits for SFLL vs. the 64 bits for Anti-SAT.

We investigated how logic locking affects the resilience of cryptographic circuits against side-channel attacks. Our results show that using logic locking schemes can potentially lead to the leakage of sensitive data in such circuits. Countermeasures such as masking can be implemented to prevent the leakage. If the high t-values are still observed even when masking is applied, this would further support our analysis. However, additional research is required to determine the effectiveness of these countermeasures in mitigating the issue.

6. LEDA: Attack Success Analysis

In this section, we explored why the proposed attacks are effective and under what conditions they produce the best results. We started by examining the pre-SAT RLL methods and their relationship with fault attacks. Both the considered methods induce perturbations into a cryptographic circuit. Our findings lead us to a constructive strategy for selecting the best incorrect locking key for our LEDFA attacks. All of the previous results were obtained using random-logic-locking keys. Following that, we looked into why LEDA performs well on circuits locked by post-SAT schemes compared to circuits locked using RLL. We have identified that the propagation of cryptographic key bits through the locking subcircuits added during logic locking is a crucial factor for successful attacks.

6.1. Locking Key Propagation for RLL

RLL inserts key gates at multiple locations within a circuit. Applying an incorrect locking key bit lk_i to any of the new inputs has a similar effect as a fault injection. While locking-enabled perturbation is easier to control, it is also more challenging than a fault attack at the same time. The equipment used for fault injection has limited spatial and temporal precision. Hence, an adversary who wants to inject a fault at a specific location may end up injecting a fault in another location or no fault at all. In contrast, by applying an incorrect locking key bit (or multiple bits), the corresponding key gate outputs will flip with certainty and perfect repeatability. On the other hand, perturbations on a desired circuit location are only possible if there is a locking key gate at that location. Even if a fault attack is unsuccessful, it can be repeated until it succeeds. However, the effects of locking circuitry are entirely deterministic.

In practice, an adversary can use the known fault attacks to apply LEDA. For instance, a well-known single-fault attack against AES-128 [14] can only work if the flipped bits are limited to one byte in the cipher state after round 7. If there are faults elsewhere in the circuit, this attack would be invalidated. Since locking can easily flip multiple key gates, it is essential to consider the relationship between such simultaneous perturbations. Figure 7 shows how two simultaneous bit flips can cancel out the effects of each other altogether, resulting in an unperturbed ciphertext that is useless for LEDA. The figure shows a part of "AES-64" that has locking key bits lk'_0 and lk'_1 driving key gates at the round 8 SBox input and ShiftRows output, respectively. If incorrect values are applied to both these locking key bits, the second induced change will fully compensate for the first one.



Figure 7. Propagation of wrong logic-locking key bits over two AES operations. The introduced errors are shown in red to demonstrate the effect of a wrong logic-locking key bit canceling out the effect of another.

In general, it is unlikely for multiple incorrect logic-locking key bits to cancel each other out due to the nature of cryptographic schemes and their inherent diffusion properties. For example, with AES variants, it is improbable that an incorrect locking key bit will be fully canceled out after a MixColumns operation, as the effects are then propagated throughout an entire column of the intermediate state. The larger the operations, the lower the probability of this effect, but it cannot be ruled out entirely. However, from the attacker's perspective, this is not a significant issue since the logic-locking key is a primary input of the circuit and can be easily changed for another incorrect logic key. In this work, the lower success rate of some LEDA attacks might be explained by the fact that the logic-locking keys were randomly selected.

On the other hand, let us consider the impact of the locking key bits placed in the most favorable positions for an adversary who can control the locking key bit values. As mentioned earlier, an incorrect cryptographic operation will affect the subsequent operations in the same way as a fault injection does. For example, suppose a collection of logic-locking key bits affects the operations of round 29 of LED. In that case, it will produce an incorrect intermediate output for LED round 29, which will impact the succeeding operations, similar to a fault injection at the start of round 30 of LED [16].

We analyzed the gate-level netlist of an LED implementation that was logic locked using an 80-bit RLL- \land locking key $lk = (lk_0, ..., lk_{79})$. Our analysis led us to identify five locking key bits $(lk_{11}, lk_{24}, lk_{27}, lk_{47}, lk_{62})$ that affect the operation of round 29 of LED. Suppose an adversary can control the locking key bits as primary inputs of the circuit. It implies they can easily choose a specific incorrect locking key lk' that flips the five locking key bits mentioned above. In our experiment, we performed one correct encryption and two incorrect encryptions using the correct locking key lk and two incorrect locking keys lk' and lk'', respectively. Finally, we solved Equation (4) (a special case of Equation (3)) in order to obtain the cryptographic key k.

$$(\operatorname{enc}_{l}(p,k,lk) = C) \wedge (\operatorname{enc}_{l}(p,k,lk') = C_{1})$$
$$\wedge (\operatorname{enc}_{l}(p,k,lk'') = C_{2})$$
(4)

Table 6 presents the results of running LEDA attacks on this logic-locked LED implementation with specific incorrect locking keys. For comparison, the table also includes the solve time and success rate of experiments with random incorrect locking keys from Table 5. These results indicate that the success rate has improved from roughly one quarter to perfect. Moreover, while the average run time improvement may not seem significant, it includes instances that were hard to solve and timed out for random locking keys but were solved successfully for the handcrafted locking keys.

Table 6. Performance of LEDA on LED-64 locked with 80-bit RLL- \land locking key. The difference from the correct locking key, lk, is shown in red.

lk =	Rando	om lk			
lk'	lk″	Solve Time [s]	Success Rate [%]	Solve Time [s]	Success Rate [%]
AB1AA45975508EBD8E8D AB1AE45975508EBD868D	AB1AE459F5508EBD8E8D AB1AA459F55087BD8E8D	725.398 784.163	100 100	869.34	26.7

6.2. Cryptographic Key Bit Propagation in Post-SAT Locking

To explain the better performance of LEDA for post-SAT methods in Table 5, we first note that they tend to introduce only one source of perturbation into the circuit. This eliminates the interference between different key gates, as discussed previously for RLL, and prevents the cancellation effect from occurring. The source is driven by the locking subcircuit, which is composed of functions g and \overline{g} for Anti-SAT and the restore unit for SFLL, as shown in Figure 3. As explained in Section 2.1.2, the inputs to this subcircuit are

known as the "tree input set", and the location where the subcircuit is inserted into the locked circuits is referred to as the "root". When the tree input set is randomly determined, the algorithm tends to insert the root closer to the end of the circuit. A parallel can be drawn between LEDA and fault attacks since the substitution–permutation networks are most vulnerable to fault injections in their last 2–3 rounds. Therefore, post-SAT locking techniques have a slightly higher chance of introducing a well-controllable perturbation in a location that is critical for cryptanalysis.

A further explanation is that during cryptanalysis, the locking subcircuit plays an important role. It is simpler and much smaller compared to the actual cryptographic circuit. For instance, in the case of Anti-SAT with a 64-bit locking key, a logic-locking block consisting of only 67 gates is inserted into an LED implementation that otherwise has 26,104 gates. The subcircuit has 32 inputs selected from the primary inputs of the circuit, which include bits of the plaintext and the cryptographic key. This creates a new path between selected bits of the cryptographic key and the circuit outputs passing through the locking subcircuit, the root and the remaining logic between the root and the outputs. This path will usually be much shorter than the existing paths between the cryptographic key and the outputs, and it goes through circuitry that was not designed with any cryptographic properties in mind. LEDA incorporates the new path in the CNF, and as a result, the SAT solver will be more effective.

In order to test our hypothesis, we ran NEOS multiple times to generate different locked circuits for Anti-SAT and SFLL. We used a 64-bit and a 128-bit locking key for both schemes. It is important to note that we deviated from our threat model in this experiment— the adversary was given a locked circuit and did not have the ability to influence the location where the locking circuitry would be added. For each run, we determined the number of cryptographic key bits in the tree input set, denoted by #CKB. The results are presented in Table 7. We discovered a clear trend: cases with more cryptographic key bits in the tree input set, while cases with only a few #CKB timed out. In some cases, the run time of successful attacks increased exponentially with #CKB. This is intuitive if we assume that LEDA can easily recover the key bits from the tree input set, and the remaining bits must be brute-forced. However, this effect is not universal; there may still be cryptographic key bits in the tree input set that are difficult to recover.

Cipher		Ant	ti-SAT			9	SFLL		
-	64-bit		1	l28-bit		64-bit			
	#CKB	Time [s]	#CKB	Time [s]	#CKB	Time [s]	#CKB	Time [s]	
	8	_	11	_	0	_	0	-	
PRESENT	13	_	24	1839.27	2	-	23	-	
5 incorrect lk	19	_	33	493.81	4	_	45	837.32	
	29	824.25	51	105.34	23	1273.85	56	211.68	
	6	2780.63	15	1977.18	0	_	3	_	
LED	12	2295.21	31	116.75	5	_	16	-	
2 incorrect lk	19	1023.20	34	89.12	12	_	21	_	
	25	582.43	57	23.47	23	1294.32	48	289.12	
	2	_	12	_	0	_	8	-	
AES-128	13	_	23	-	10	_	4	-	
2 incorrect lk	22	_	31	-	27	_	29	_	
	30	-	64	8563.64	42	14,681.83	65	10,426.53	

Table 7. Numbers **#CKB** of cryptographic key bits among the tree input set and LEDA run times in seconds ("–" indicates a timeout).

One notable aspect that stands out from the findings presented in Table 7 is the remarkable effectiveness of LEDA against SFLL, which previously exhibited poor results as demonstrated in Table 5. When comparing Anti-SAT and SFLL, we observe similar timeout behavior and run times for comparable #CKB. However, in this case, LEDA appears also

to be effective against SFLL, particularly when a sufficient amount of #CKB is present. The vital distinction between Anti-SAT and SFLL is that the former only allows the primary inputs of the circuit in the tree input set, while the latter permits intermediate wires as well. Consequently, the likelihood of encountering a cryptographic key bit among the primary inputs is significantly higher than among all the signals, resulting in a much higher average #CKB for Anti-SAT. Therefore, we can conclude that the ineffective performance of LEDA against SFLL, as shown in Table 5, was most likely due to a low #CKB in randomly generated locking instances.

7. Proposed Countermeasure for LEDA

In Section 6.2, we evaluate attack success rates for LEDA, and based on it, we propose a strategy to prevent the propagation of cryptographic key bits to the primary outputs through the smaller locking subcircuits. The results of our experiments, presented in Table 7, indicate that the success rate of LEDA is highly dependent on the presence of cryptographic key bits in the tree input set. To achieve better security levels, circuit designers utilizing logic locking schemes for cryptographic circuits must carefully consider the placement of cryptographic key bits and avoid introducing them directly into the locking subcircuits.

In order to evaluate the effectiveness of our proposed method, we conducted a series of experiments where we modified the post-SAT logic locking schemes in NEOS. Specifically, we excluded cryptographic key bits from the set of eligible primary inputs for the tree input set. Instead, we derived the tree input set exclusively from plaintext bits or intermediate wires. We then generated a set of locked cryptographic circuits using Anti-SAT and SFLL schemes. We verified that none of these circuits contained any cryptographic key bits within the tree input set.

Afterwards, we proceeded to conduct LEDA attacks on each circuit and compared the results to those obtained from previous experiments. As indicated in Table 8, our proposed countermeasure proved to be highly effective, reducing the success rate to nearly zero in most cases. These results are in stark contrast to the 100% success rate observed in Table 7. However, we did observe a few successful attacks in certain scenarios, such as five successful attacks on PRESENT locked using a 64-bit Anti-SAT locking key when the attack formula was constructed using 10 incorrect logic-locking keys.

Cipher			An	ti-SAT		SFLL					
	#Incorrect	6	64-bit		28-bit	6	4-bit	128-bit			
	Locking	Success	Solve	Success	Solve	Success	Solve	Success	Solve		
	Keys	Rate	Time [s]	Rate	Time [s]	Rate	Time [s]	Rate	Time [s]		
DDECENIT	10	0.167	10,284.62	0.033	16,579.37	0	-	0	_		
I KESEN I	5	0	_	0	-	0	-	0	-		
LED	2	0.033	8271.09	0.067	11,843.30	0	-	0	_		
"AES-64"	2	0	_	0.033	17,542.89	0	-	0	_		
AES-128	2	0	_	0	_	0	-	0	-		

Table 8. Success rates and average LEDA run times in seconds for successful attacks on circuits with modified logic locking schemes, 30 repetitions ("–" indicates a timeout).

Furthermore, we also noted that the average solve time for these successful attacks was significantly longer compared to the results observed in both Tables 5 and 7. Overall, these findings suggest that our proposed method is a highly effective countermeasure against LEDA attacks, although further improvements may still be possible in specific scenarios.

8. Conclusions and Future Work

We proposed two attacks that utilize the presence of logic-locking circuitry for faultbased (LEDFA) and classical (LEDA) cryptanalysis. The reason the attacks are effective is that the improperly unlocked circuits retain access to the cryptographic key while implementing functions that may have less cryptographic strength. Furthermore, new propagation paths that are introduced by the locking circuitry can make cryptanalysis significantly simpler. The attacks continued to work in our experiments even when random selections of incorrect locking keys were made. The success rate dramatically increased when optimized locking keys that caused disturbances in the vicinity of circuit locations susceptible to known fault attacks were used. We suggest prohibiting the direct feeding of cryptographic key bits into locking subcircuits. Our findings demonstrate how the circuits locked using the modified logic locking schemes are much more resistant to LEDA.

In addition, we investigated side-channel leakage of multiple logic-locked AES implementations. This first assessment using TVLA methodology, as shown in Section 5.3.3, showcases interesting leakage patterns compared to a standard AES implementation. Our results suggest that locking has a non-negligible influence on information leaked from the implementation. Dedicated SCA countermeasures are necessary in the case of locked and unlocked cryptographic circuits alike. Such protections must be verified to be effective under different locking keys. This is, however, out of scope for this paper, but investigating protected circuits would be interesting as a continuation of this work.

Despite not being created with cryptographic circuits in mind, no previous work has explicitly excluded logic-locking techniques from its purview. Because of this, critical system designers might wish to use logic locking to add an extra degree of security, which, unfortunately, would weaken the systems. This work is the first to use quantitative analysis to both identify and validate the vulnerability introduced by such schemes. The results of this work suggest that using logic locking in cryptographic circuits is not safe. The designer must avoid adding key gates to crucial locations where fault attacks could occur (e.g., the last 3–4 rounds in substitution–permutation networks). Furthermore, future locking schemes can be expanded with the ability to identify an extended use of incorrect locking keys, in a manner similar to identifying fault attacks or attempted tampering.

Author Contributions: Conceptualization, D.U., M.G. and I.P.; investigation, D.U.; writing—original draft preparation, D.U.; writing—review and editing, D.U., M.G. and I.P.; supervision, I.P. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The data presented in this study are available in article.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Bogdanov, A.; Knudsen, L.R.; Leander, G.; Paar, C.; Poschmann, A.; Robshaw, M.J.; Seurin, Y.; Vikkelsoe, C. PRESENT: An ultralightweight block cipher. In Proceedings of the Cryptographic Hardware and Embedded Systems-CHES 2007: 9th International Workshop, Vienna, Austria, 10–13 September 2007; Proceedings 9; Springer: Berlin/Heidelberg, Germany, 2007; pp. 450–466.
- 2. Rostami, M.; Koushanfar, F.; Karri, R. A primer on hardware security: Models, methods, and metrics. *Proc. IEEE* 2014, 102, 1283–1295. [CrossRef]
- Knechtel, J.; Kavun, E.B.; Regazzoni, F.; Heuser, A.; Chattopadhyay, A.; Mukhopadhyay, D.; Dey, S.; Fei, Y.; Belenky, Y.; Levi, I.; et al. Towards Secure Composition of Integrated Circuits and Electronic Systems: On the Role of EDA. In Proceedings of the Design, Automation & Test in Europe Conference (DATE), Grenoble, France, 9–13 March 2020; pp. 508–513. [CrossRef]
- 4. Regazzoni, F.; Breveglieri, L.; Ienne, P.; Koren, I. Interaction between fault attack countermeasures and the resistance against power analysis attacks. In *Fault Analysis in Cryptography*; Springer: Berlin/Heidelberg, Germany, 2012; pp. 257–272.
- 5. Joye, M. Security paradoxes: How improving a cryptosystem may weaken it. In Proceedings of the National Conference on Information Security, Arlington, VA, USA, 18–21 October 1999; pp. 27–32.
- Joye, M.; Quisquater, J.J.; Sung-Ming, Y.; Yung, M. Observability analysis-detecting when improved cryptosystems fail. In Proceedings of the Cryptographers' Track at the RSA Conference, San Jose, CA, USA, 18–22 February 2002; Springer: Berlin/Heidelberg, Germany, 2002; pp. 17–29.

- 7. Subramanyan, P.; Ray, S.; Malik, S. Evaluating the security of logic encryption algorithms. In Proceedings of the IEEE International Symposium on Hardware Oriented Security and Trust (HOST), Washington, DC, USA, 5–7 May 2015; pp. 137–143.
- Alrahis, L.; Patnaik, S.; Khalid, F.; Hanif, M.A.; Saleh, H.H.; Shafique, M.; Sinanoglu, O. GNNUnlock: Graph Neural Networksbased Oracle-less Unlocking Scheme for Provably Secure Logic Locking. In Proceedings of the Design, Automation & Test in Europe Conference, (DATE), Grenoble, France, 1–5 February 2021; pp. 780–785. [CrossRef]
- 9. Yasin, M.; Mazumdar, B.; Rajendran, J.J.; Sinanoglu, O. SARLock: SAT attack resistant logic locking. In Proceedings of the IEEE International Symposium on Hardware Oriented Security and Trust (HOST), McLean, VA, USA, 3–5 May 2016; pp. 236–241.
- Xie, Y.; Srivastava, A. Mitigating SAT attack on logic locking. In Proceedings of the Cryptographic Hardware and Embedded Systems–CHES 2016: 18th International Conference, Santa Barbara, CA, USA, 17–19 August 2016; Proceedings 18; Springer: Berlin/Heidelberg, Germany, 2016; pp. 127–146.
- Yasin, M.; Sengupta, A.; Schafer, B.C.; Makris, Y.; Sinanoglu, O.; Rajendran, J. What to lock? Functional and parametric locking. In Proceedings of the Great Lakes Symposium on VLSI, Banff, AB, Canada, 10–12 May 2017; pp. 351–356.
- Yasin, M.; Sengupta, A.; Nabeel, M.T.; Ashraf, M.; Rajendran, J.; Sinanoglu, O. Provably secure logic locking: From theory to practice. In Proceedings of the 2017 ACM SIGSAC Conference on Computer and Communications Security, Dallas, TX, USA, 30 October–3 November 2017; pp. 1601–1618.
- Upadhyaya, D.; Gay, M.; Polian, I. LEDA: Locking Enabled Differential Analysis of Cryptographic Circuits. In Proceedings of the 2023 IEEE International Symposium on Hardware Oriented Security and Trust (HOST), San Jose, CA, USA, 1–4 May 2023; pp. 249–259.
- Tunstall, M.; Mukhopadhyay, D.; Ali, S. Differential fault analysis of the advanced encryption standard using a single fault. In Proceedings of the IFIP International Workshop on Information Security Theory and Practices, Heraklion, Greece, 1–3 June 2011; Springer: Berlin/Heidelberg, Germany, 2011; pp. 224–233.
- Piret, G.; Quisquater, J.J. A differential fault attack technique against SPN structures, with application to the AES and KHAZAD. In Proceedings of the International Workshop on Cryptographic Hardware and Embedded Systems, Cologne, Germany, 8–10 September 2003; Springer: Berlin/Heidelberg, Germany, 2003; pp. 77–88.
- Jovanovic, P.; Kreuzer, M.; Polian, I. A fault attack on the LED block cipher. In Proceedings of the Constructive Side-Channel Analysis and Secure Design: Third International Workshop, COSADE 2012, Darmstadt, Germany, 3–4 May 2012; Proceedings 3; Springer: Berlin/Heidelberg, Germany, 2012; pp. 120–134.
- 17. Park, S.H.; Jeong, K.T.; Lee, Y.S.; Sung, J.C.; Hong, S.H. Improved Differential Fault Analysis on Block Cipher PRESENT-80/128. *J. Korea Inst. Inf. Secur. Cryptol.* **2012**, 22, 33–41.
- 18. Zhang, F.; Guo, S.; Zhao, X.; Wang, T.; Yang, J.; Standaert, F.X.; Gu, D. A framework for the analysis and evaluation of algebraic fault attacks on lightweight block ciphers. *IEEE Trans. Inf. Forensics Secur.* **2016**, *11*, 1039–1054. [CrossRef]
- 19. Bhunia, S.; Hsiao, M.S.; Banga, M.; Narasimhan, S. Hardware Trojan attacks: Threat analysis and countermeasures. *Proc. IEEE* **2014**, *102*, 1229–1247. [CrossRef]
- Guin, U.; Huang, K.; DiMase, D.; Carulli, J.M.; Tehranipoor, M.; Makris, Y. Counterfeit integrated circuits: A rising threat in the global semiconductor supply chain. *Proc. IEEE* 2014, 102, 1207–1228. [CrossRef]
- 21. Rajendran, J.; Sam, M.; Sinanoglu, O.; Karri, R. Security analysis of integrated circuit camouflaging. In Proceedings of the 2013 ACM SIGSAC Conference on Computer & Communications Security, Berlin, Germany, 4–8 November 2013; pp. 709–720.
- 22. Yasin, M.; Rajendran, J.J.; Sinanoglu, O.; Karri, R. On improving the security of logic locking. *IEEE Trans. Comput.-Aided Des. Integr. Syst.* 2015, 35, 1411–1424. [CrossRef]
- Kahng, A.B.; Lach, J.; Mangione-Smith, W.H.; Mantik, S.; Markov, I.L.; Potkonjak, M.; Tucker, P.; Wang, H.; Wolfe, G. Watermarking techniques for intellectual property protection. In Proceedings of the 35th annual Design Automation Conference, San Francisco, CA, USA, 15–19 June 1998; pp. 776–781.
- 24. Caldwell, A.E.; Choi, H.J.; Kahng, A.B.; Mantik, S.; Potkonjak, M.; Qu, G.; Wong, J.L. Effective iterative techniques for fingerprinting design IP. *IEEE Trans. Comput.-Aided Des. Integr. Syst.* 2004, 23, 208–215. [CrossRef]
- 25. Alkabani, Y.; Koushanfar, F. Active Hardware Metering for Intellectual Property Protection and Security. In Proceedings of the USENIX Security Symposium, Boston, MA, USA, 6–10 August 2007; Volume 20, pp. 1–20.
- 26. Roy, J.A.; Koushanfar, F.; Markov, I.L. Ending piracy of integrated circuits. Computer 2010, 43, 30–38. [CrossRef]
- 27. Rajendran, J.; Pino, Y.; Sinanoglu, O.; Karri, R. Security analysis of logic obfuscation. In Proceedings of the 49th Annual Design Automation Conference, San Francisco, CA, USA, 3–7 June 2012; pp. 83–89.
- 28. Rajendran, J.; Pino, Y.; Sinanoglu, O.; Karri, R. Logic encryption: A fault analysis perspective. In Proceedings of the 2012 Design, Automation & Test in Europe Conference & Exhibition (DATE), Dresden, Germany, 12–16 March 2012; pp. 953–958.
- 29. Baumgarten, A.; Tyagi, A.; Zambreno, J. Preventing IC piracy using reconfigurable logic barriers. *IEEE Des. Test Comput.* **2010**, 27, 66–75. [CrossRef]
- Shakya, B.; Xu, X.; Tehranipoor, M.; Forte, D. Cas-lock: A security-corruptibility trade-off resilient logic locking scheme. *IACR Trans. Cryptogr. Hardw. Embed.* 2020, 2020, 175–202. [CrossRef]
- Shamsi, K.; Li, M.; Meade, T.; Zhao, Z.; Pan, D.Z.; Jin, Y. Cyclic obfuscation for creating SAT-unresolvable circuits. In Proceedings of the on Great Lakes Symposium on VLSI 2017, Banff, AB, Canada, 10–12 May 2017; pp. 173–178.
- 32. Yasin, M.; Rajendran, J.J.; Sinanoglu, O. A brief history of logic locking. In *Trustworthy Hardware Design: Combinational Logic Locking Techniques*; Springer: Berlin/Heidelberg, Germany, 2020; pp. 17–31.

- Bar-El, H.; Choukri, H.; Naccache, D.; Tunstall, M.; Whelan, C. The sorcerer's apprentice guide to fault attacks. *Proc. IEEE* 2006, 94, 370–382. [CrossRef]
- Van Woudenberg, J.G.; Witteman, M.F.; Menarini, F. Practical optical fault injection on secure microcontrollers. In Proceedings of the 2011 Workshop on Fault Diagnosis and Tolerance in Cryptography, Tokyo, Japan, 29 September 2011; pp. 91–99.
- Dehbaoui, A.; Dutertre, J.M.; Robisson, B.; Tria, A. Electromagnetic transient faults injection on a hardware and a software implementations of AES. In Proceedings of the 2012 Workshop on Fault Diagnosis and Tolerance in Cryptography, Leuven Belgium, 9 September 2012; pp. 7–15.
- Li, Y.; Sakiyama, K.; Gomisawa, S.; Fukunaga, T.; Takahashi, J.; Ohta, K. Fault sensitivity analysis. In Proceedings of the Cryptographic Hardware and Embedded Systems, CHES 2010: 12th International Workshop, Santa Barbara, CA, USA, 17–20 August 2010; Proceedings 12; Springer: Berlin/Heidelberg, Germany, 2010; pp. 320–334.
- Ghalaty, N.F.; Yuce, B.; Taha, M.; Schaumont, P. Differential fault intensity analysis. In Proceedings of the 2014 Workshop on Fault Diagnosis and Tolerance in Cryptography, Busan, Republic of Korea, 23 September 2014; pp. 49–58.
- Gay, M.; Paxian, T.; Upadhyaya, D.; Becker, B.; Polian, I. Hardware-oriented algebraic fault attack framework with multiple fault injection support. In Proceedings of the 2019 Workshop on Fault Diagnosis and Tolerance in Cryptography (FDTC), Atlanta, GA, USA, 24 August 2019; pp. 25–32.
- 39. Kocher, P.; Jaffe, J.; Jun, B. Introduction to Differential Power Analysis and Related Attacks. 1998. Available online: https://www.rambus.com/wp-content/uploads/2015/08/DPATechInfo.pdf (accessed on 15 August 2023).
- Becker, G.; Cooper, J.; DeMulder, E.; Goodwill, G.; Jaffe, J.; Kenworthy, G.; Kouzminov, T.; Leiserson, A.; Marson, M.; Rohatgi, P.; et al. Test vector leakage assessment (TVLA) methodology in practice. In Proceedings of the International Cryptographic Module Conference, Gaithersburg, MD, USA, 24–26 September 2013; Volume 1001, p. 13.
- 41. Schneider, T.; Moradi, A. Leakage assessment methodology: Extended version. J. Cryptogr. Eng. 2016, 6, 85–99. [CrossRef]
- 42. Papagiannopoulos, K.; Glamočanin, O.; Azouaoui, M.; Ros, D.; Regazzoni, F.; Stojilović, M. The side-channel metrics cheat sheet. *ACM Comput. Surv.* **2023**, *55*, 1–38. [CrossRef]
- 43. Biham, E.; Shamir, A. Differential cryptanalysis of DES-like cryptosystems. J. Cryptol. 1991, 4, 3–72. [CrossRef]
- Torrance, R.; James, D. The state-of-the-art in IC reverse engineering. In Proceedings of the International Workshop on Cryptographic Hardware and Embedded Systems, Lausanne, Switzerland, 6–9 September 2009; Springer: Berlin/Heidelberg, Germany, 2009; pp. 363–381.
- 45. Tseitin, G.S. On the complexity of derivation in propositional calculus. In *Automation of Reasoning*; Springer: Berlin/Heidelberg, Germany, 1983; pp. 466–483.
- Shamsi, K.; Yier, J. Netlist Encryption and Obfuscation Suite. 2019. Available online: https://bitbucket.org/kavehshm/neos/src/master/ (accessed on 25 September 2021).
- Soos, M.; Nohl, K.; Castelluccia, C. Extending SAT solvers to cryptographic problems. In Proceedings of the International Conference on Theory and Applications of Satisfiability Testing, Swansea, UK, 30 June–3 July 2009; Springer: Berlin/Heidelberg, Germany, 2009; pp. 244–257.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.